Author's response to reviews

Title: Implications for Registry-Based Vaccine Effectiveness Studies from an Evaluation of an Immunization Registry: A Cross-sectional Study

Authors:

Barbara E Mahon (mahonbe@bu.edu)
Kimberly M Shea (kimshea@bu.edu)
Nancy N Dougherty (nancy.dougherty@novartis.com)
Anita M Loughlin (aloughli@bu.edu)

Version: 3 Date: 30 April 2008

Author's response to reviews:

We have made the formatting changes requested, including:
1. Using sentence case in the title
2. Removing the authors' qualifications
3. Including the country in the affiliation details
4. Removing the mailing addresses